Author’s response to reviews

Title: Persistent chronic hypokalemia in a patient with anorexia nervosa - diagnostic and treatment dilemmas

Authors:

Kevin Mohee (kevin.mohee@nhs.net)
Kate Kucharska-Pietura (kate.kucharska-pietura@nhs.net)
Amalan Karthigeyan (amalan.karthigeyan@nhs.net)
Asif Naqvi (asifabbas.naqvi@nhs.net)

Version: 2  Date: 1 April 2014

Author's response to reviews: see over
Dear Editor

We believe that this a unique case report unpublished before because it highlights how diagnosis and treatment represent a real dilemma for clinicians from different specialties involved in this patient’s care. We carried out a number of investigations including advanced neuropsychological assessments to help us better understand the diagnosis and plan treatment. In addition to that, a comprehensive review of all the tests involved in a patient with hypokalaemia and we explained the rationale of each tests as well as we provided relevant differential diagnoses. We believe that this case report will undoubtedly help in our understanding of hypokalaemia in anorexia nervosa patients. It will have a broader clinical impact in addition to psychiatry and gastroenterology.

Dr Kevin Mohee